Author’s response to reviews

Title: The effects of temperature on nitrous oxide and oxygen mixture homogeneity and stability

Authors:

Patrick D Litwin (plitwin@telus.net)

Version: 11 Date: 14 October 2010

Author’s response to reviews:

There was one sentence at the end of the first paragraph of the background section of the paper which earlier was deleted in agreement with the reviewers but was inadvertently reinserted in a subsequent submission. I have removed it again from the final version.